Author's response to reviews

Title: Dietary supplementation with rice bran fermented with Lentinus edodes increases interferon-gamma activity without causing adverse effects: a randomized, double-blind, placebo-controlled, parallel-group study

Authors:

Ji-young Choi (cci55@nate.com)
Doo-Jin Paik (paikdj@hanyang.ac.kr)
Dae Young Kwon (dykwon@kfri.re.kr)
Yongsoon Park (Yongsoon@hanyang.ac.kr)

Version: 4 Date: 4 April 2014

Author's response to reviews:

Dear Editor-in-chief,

We submit the revised manuscript, “MS: 4866959281186486 - Supplementation of rice bran fermented with Lentinus edodes increases interferon-γ activities without causing adverse effects: a randomized, double-blind, placebo-controlled, parallel-group study” for publication in Nutrition Journal.

Referee 1 had no comment. According to the editorial comments, we provide P values in tables, and our manuscript are formatted correctly.

Sincerely,
Yongsoon Park
Department of Food and Nutrition, Hanyang University
222 Wangsimni-ro, Seondong-gu, Seoul, 133-791, South Korea
E-mail: Yongsoon@hanyang.ac.kr
Tel: +82-02-2220-1205; Fax: +82-02-2220-1856